Author’s response to reviews

Title: Birth-weight, insulin levels, and HOMA-IR in newborns at term.

Authors:

Luis E Simental-Mendía (luis_simental81@hotmail.com)
Argelia Castañeda-Chacón (dra_argelia_chacon@hotmail.com)
Martha Rodríguez-Morán (rodriguez_moran@hotmail.com)
Fernando Guerrero-Romero (guerrero_romero@hotmail.com)

Version: 3 Date: 4 May 2012

Author’s response to reviews:

Editorial comment:
The revised manuscript is very interesting and we look forward to seeing this in publication. The following sentence in the discussion section is confusing as written. Please rewrite. The confusion arises from the section disagree that could be related?? "However, insulin levels differed among weight classes, which in the study by Wolf et al. [15] was in the sequence of SGA<AGA<LGA whereas in our study in the sequence of AGA<SGA<LGA; disagree that could be related with differences in the birth-weight of LGA (3824 ± 364 g versus 4175 ± 197 g) and SGA (2516 ± 352 g versus 2083±356 g) newborns in the study by Wolf as compared with our results, respectively."

Response. The phrase was rewritten (Page 8, lines 4-13)

Misspelling of Abbott in the manuscript.
Response. Misspelling was corrected (Page 5, line 23)